**Author's response to reviews**

**Title:** Vitamin D levels are low in adult patients with sickle cell disease in Jamaica and West Africa

**Authors:**

Bamidele O Tayo (btayo@lumc.edu)
Titilola S Akingbola (titiakingbola@yahoo.com)
Babatunde L Salako (tundesalako@hotmail.com)
Colin A McKenzie (colin.mckenzie02@uwimona.edu.jm)
Marvin Reid (marvin.reid@uwimona.edu.jm)
Jennifer Layden (jlayden@lumc.edu)
Ifeyinwa Osunkwo (Ify.Osunkwo@choa.org)
Jacob Plange-Rhule (jprhule@africaonline.com.gh)
Amy Luke (aluke@lumc.edu)
Ramon Durazo-Arvizu (rdurazo@lumc.edu)
Richard S Cooper (rcooper@lumc.edu)

**Version:** 5  
**Date:** 1 August 2014

**Author's response to reviews:**

Statement on adherence to the STROBE guidelines for observational studies has now been included in the statistical analysis sub-section of the method section on page #6 and reference #21 included as a citation for STROBE.

Keywords are now below the abstract.

List of abbreviations used in the text is now provided after the conclusion on page #10.